Author's response to reviews

Title: A Comparison of Online versus On-site Training in Health Research
Methodology: A Randomized Study

Authors:

Rakesh Aggarwal (aggarwal.ra@gmail.com)
Nikhil Gupte (nikhil_jhumit@yahoo.com)
Nancy Kass (nkass@jhsph.edu)
Holly Taylor (htaylor@jhsph.edu)
Joseph Ali (jali@jhsph.edu)
Anant Bhan (anantbhan@gmail.com)
Amita Aggarwal (aa.amita@gmail.com)
Stephen D Sisson (ssisson@jhmi.edu)
Sukon Kanchanaraksa (skanchan@jhsph.edu)
Jane McKenzie-White (jmw@jhmi.edu)
John McGready (jmcmready@jhsph.edu)
Paolo Miotti (miottip@od.nih.gov)
Robert C Bollinger (rcb@jhmi.edu)

Version: 3 Date: 15 June 2011

Author's response to reviews:

Revisions have been done as per the email of June 1 from the journal editorial office. These include:

a. Change of the heading 'Introduction' to 'Background'.

b. A conclusion has been added.

c. Authors’ contributions: A sentence has been added to indicate that 'All the authors have read and approved the final manuscript.'.

d. The tables numbered as 3 and 4 in the previous version had 3 components with different number of columns. This posed a problem for the journal. Hence, each table has been split into 3 tables (now numbered as table 3-8).

e. It has been indicated that there are no 'Additional files'.

f. A few minor spelling and other corrections have been done.